Author's response to reviews

Title: Role of Caveolin 1, E-Cadherin, Enolase 2 and PKC alpha on resistance to methotrexate in human HT29 colon cancer cells

Authors:

   Elisabet Selga (eliselga@ub.edu)
   Cristina Morales (cristina.morales@bric.dk)
   Veronique Noe (vnoe@ub.edu)
   Miguel A Peinado (map@imppc.org)
   Carlos J Ciudad (cciudad@ub.edu)

Version: 5 Date: 1 August 2008

Author's response to reviews:

Dear BMC Editorial Production Team

Enclosed please find the formatted version of our manuscript entitled: Role of Caveolin 1, E-Cadherin, Enolase 2 and PKCa on resistance to methotrexate in human HT29 colon cancer cells. Now all authors in the references are displayed. We hope this version is OK.

Sincerely yours

Carlos J. Ciudad